Author's response to reviews

Title: Impairment of intellectual functions after surgery and posterior fossa irradiation in children with ependymoma is related to age and neurologic complications

Authors:

Katja von Hoff (hoff_k@klinik.uni-wuerzburg.de)
Virginie Kieffer (kieffer@igr.fr)
Jean-Louis Habrand (habrand@igr.fr)
Chantal Kalifa (kalifa@igr.fr)
Georges Dellatolas (dellatolas@vjf.inserm.fr)
Jacques Grill (grill@igr.fr)

Version: 3 Date: 2 October 2007

Author's response to reviews:

Please find herewith then revision of our paper.
A table has been added and the legends of the figures and tables clarified.